Author’s response to reviews

Title: What is the financial incentive to immigrate? An analysis of salary disparities between Health Workers working in the Caribbean and popular destination countries

Authors:

Gavin George (Georgeg@ukzn.ac.za)
Bruce Rhodes (Rhodesb@ukzn.ac.za)
Christine Laptiste (Christine.Laptiste@sta.uwi.edu)

Version: 2 Date: 10 Jan 2019

Author’s response to reviews:

As instructed by the editor:

Thank you for providing a response to the reviewers/Cover Letter. As these documents are no longer required at this stage of the publication process, please remove them from your submission’s supplementary files.